Author’s response to reviews

Title: Stakeholders’ perceptions of transferability criteria for health promotion interventions: a case study.

Authors:

Justine Trompette (justine.trompette@gmail.com)
Joëlle Kivits (joelle.kivits@univ-lorraine.fr)
Laetitia Minary (l.minary@chu-nancy.fr)
Linda Cambon (linda.cambon@wanadoo.fr)
François Alla (francois.alla@univ-lorraine.fr)

Version: 2 Date: 9 June 2013

Author’s response to reviews:

Dear Mrs. Audrey Ann Reyes,

Our study did not involve individual subjects or identifying health data. Thus, the regulations regarding human research are not concerned.

As suggested we added a sentence and a reference in our manuscript (page 5):

“Ethical issues. This study did not involve individual subjects or identifying health data. Thus, the regulations regarding human research are not concerned (ref). Professionals who participate gave their consent to participate.”

Best regards

Justine Trompette